Author's response to reviews

Title: Transcript levels of Toll-like receptors 5, 8 and 9 correlate with inflammatory activity in Ulcerative Colitis

Authors:

Fausto Sanchez-Muñoz (fausto22@yahoo.com)
Gabriela Fonseca-Camarillo (gabrielafaster@gmail.com)
Marco A Villeda-Ramírez (nmavilledar_snk@hotmail.com)
Elizabeth Miranda-Perez (mirandapme@yahoo.com.mx)
Edgar J Mendivil (ed-jair@hotmail.com)
Rafael Barreto-Zuñiga (barretozu@yahoo.com)
Misael Uribe (muribe@rtn.net.mx)
Rafael Bojalil (bojraf@yahoo.com)
Aarón Domínguez-Lopez (aadominguezl@yahoo.com.mx)
Jesus K Yamamoto-Furusho (kazuofurusho@hotmail.com)

Version: 7 Date: 25 November 2011

Author's response to reviews: see over
Mexico City, November 25th, 2011

Tim Shipley, PhD

BMC Gastroenterology

Editor in Chief

Dear Editor:

Enclosed you will find the third revised version of the manuscript: MS: 1380844289464145 - Transcript levels of Toll-like receptors 5, 8 and 9 correlate with inflammatory activity in Ulcerative Colitis and the response for the final revisions, already corrected in the manuscript.

Below we submit the detail response.

I am looking forward hearing from you soon.

Best regards,

Prof. Jesus K. Yamamoto-Furusho M.D., Ph.D.

Head of IBD Clinic

Department of Gastroenterology

Instituto Nacional de Ciencias Médicas y Nutrición
Major revision

Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order.

Authors' contributions: Please include the statement 'All authors read and approved the final manuscript'.

Minor revisions

Please remove any page numbers

Re: page numbers were removed.

Highlighting/tracking: Please remove all highlighting and tracking from the manuscript.

Re: manuscript is now without highlighting and tracking in black.

Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. This comment refers to Figures 1 and 2. Our online figure guide contains full details for preparing files for submission and can be viewed here: http://www.biomedcentral.com/info/ifora/figures

Re: figures have been formatted.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

Re: We revised the manuscript.